Author’s response to reviews

Title: Acute oral toxicity and antioxidant studies of an amine-based diselenide

Authors:

Mohammad Ibrahim (dribrahim@awkum.edu.pk; ibrahim189@yahoo.com)
Niaz Muhammad (drniaz@awkum.edu.pk)
Musadiq Ibrahim (musadiqibrahim@gmail.com)
Muhammad Ishaq Ali Shah (ishaqalishah@awkum.edu.pk)
Muhammad Said (msaidqau@yahoo.com)
Jean Paul Kamdem (kamdemjeanpaul2005@yahoo.fr)
Joao Batista Teixeira Rocha (jbtrocha@yahoo.com.br)
Muhammad Khan (idress189@yahoo.com)
Wali Ullah (drwali@awkum.edu.pk)

Version: 5 Date: 13 Mar 2019

Author’s response to reviews:

Dear Editor,

BMC Complementary & Alternative Medicine
BCAM-D-18-00502R4

Thank you for your useful comments and suggestions on the structure of our manuscript. We are submitting our revised R4 manuscript entitled “Acute Liver toxicity tests and antioxidant profile of an amine-based diselenide”. We have modified the manuscript accordingly, and the detailed corrections are listed below point by point:

We note that the current submission contains some textual overlap with other previously published works, in particular:
This overlap exists throughout your manuscript. While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work.

Thank you for your suggestion, now the textual overlap has been significantly removed.

Dr. Mohammad Ibrahim

Department of Chemistry,

Abdul Wali Khan University Mardan,

Pakistan.

E-mail: Ibrahim189@yahoo.com; dribrahim@awkum.edu.pk